Author’s response to reviews

Title: Treatment in the home setting with Intermittent Pneumatic Compression for a woman with chronic leg ulcers: A Case Report

Authors:
Katrina Young (katrina.young@health.nsw.gov.au)
Harrison Ng Chok (h.ngchok@westernsydney.edu.au)
Lesley Wilkes (l.wilkes@uws.edu.au)

Version: 1 Date: 01 May 2017

Author’s response to reviews:

Dear Reviewer,

Please see the following changes have been made to the manuscript submission

1) Please see our submission guidelines for information on how to correctly prepare your Case Report. A full Declarations list is required at the end of your manuscript, please see the link below for further advice on how to correctly complete each section.

https://bmcnurs.biomedcentral.com/submission-guidelines/preparing-your-manuscript/case-report.

The following was added to the end of the manuscript

DECLARATIONS

ETHICS APPROVAL AND CONSENT TO PARTICIPATE

Ethics approval was granted for the Clinical Case Report and Media and Image use from the Nepean Blue Mountains Local Health District Human Research Ethics Office. The authors would like to confirm that all treatments presented within this Case Report are part of standard care. The client also provided informed consent to participate.

CONSENT TO PUBLISH
A media and image consent form along detailing the consent to publish was granted on the 10/05/16 by the patient.

AVAILABILITY OF DATA AND MATERIALS
Consent for publication of raw photos and data obtained from study participant

COMPETING INTERESTS
The authors declare no competing interests

FUNDING
This project was not funded

AUTHORS CONTRIBUTIONS
KY supported the case client in this case report and reported the findings. KY and HNC was involved in the literature review. KY, HNC and LW were involved in the editing and proofing of draft manuscript of this case report. All authors read and approved the final manuscript.

ACKNOWLEDGEMENTS
Acknowledgement section woman/client for allowing us to share her story, Nepean Blue Mountains Local Health District (NBMLHD), Clinical Nursing Research and Practice Department (CNRPD).

2) In your "Ethics and Consent to Participate" declaration, please confirm whether all treatments presented within this Case Report are part of standard care.

Added to section.
3) Please note, we require written consent from the patient to publish their individual information and images. Please note, the Editor may ask to see this consent form at any given time.

Available from first author upon request from the editors at any time.

4) Please upload a completed CARE checklist with your revised manuscript.

Added to the manuscript/

Kind regards,

Authors